Author's response to reviews

Title: LEIGC long non-coding RNA acts as a tumor suppressor in gastric carcinoma by inhibiting the epithelial-to-mesenchymal transition

Authors:

Yuehua Han (hanyh_2012@aliyun.com)
Jun Ye (wzmcyejun@zju.edu.cn)
Dang Wu (dangwuzju@126.com)
Pin Wu (doctorwp@163.com)
Zhigang Chen (pridechen@126.com)
Jian Chen (ChenJ@126.com)
Shuanliang Gao (Gaosl@163.com)
Jian Huang (Drhuangjian@zju.edu.cn)

Version: 6 Date: 19 November 2014

Author's response to reviews: see over
Dear editor,

Please consider a revised version of our manuscript (MS: 1044206819132426) entitled “LEIGC long non-coding RNA acts as a tumor suppressor in gastric carcinoma by inhibiting the epithelial-to-mesenchymal transition”.

Thank you and the reviewers for the important suggestions. We revised our manuscript according to editor's request. We hope that our revised manuscript will meet your journal requirement.

Our point by point response is as follows.

Best regards,

Jian Huang
Point by Point Response

1. We included the name of the animal ethics committee which approved this study at line 193-195.

2. We removed our conflict of interest declaration from additional files.

3. We included figure legend for supplementary figure after the figure legends section and removed it from the additional files.